Author's response to reviews

Title: Assessing communities of practice in health policy: A conceptual framework as a first step towards empirical research

Authors:

Maria Paola Bertone (mpbertone@yahoo.com)
Bruno Meessen (bmeessen@itg.be)
Guy Clarysse (gclarysse@unicef.org)
David Hercot (dhercot@itg.be)
Allison Kelley (agkelley514@yahoo.com)
Yamba Kafando (odnafak@yahoo.fr)
Isabelle Lange (Isabelle.Lange@lshtm.ac.uk)
Jérôme Pfaffmann (jpfaeffmann@unicef.org)
Valéry Ridde (valery.ridde@umontreal.ca)
Isidore Sieleunou (isidoros2002@yahoo.fr)
Sophie Witter (s.witter@abdn.ac.uk)

Version: 3 Date: 3 October 2013

Author's response to reviews: see over
"Assessing communities of practice in health policy: A conceptual framework as a first step towards empirical research"
A paper submitted to Health Research Policy and Systems

Dear Editors,

Please find attached the revised version of our paper entitled “Assessing communities of practice in health policy: A conceptual framework as a first step towards empirical research”.

Following the latest comments from our reviewers, we have corrected a minor typo and we have cut out most the section on the “application of the conceptual framework”, where it was making general remarks and not referring specifically to the proposed framework.
A final proof-reading of the paper was done to correct all typos and minor mistakes in the text as well as in the table, figure and additional file.

Once again, we would like to express our gratitude to the reviewers for their excellent work and their help in improving our paper. We also very much appreciate your time and consideration and look forward to hearing from you soon.

With best regards,

the authors